Author's Response To Reviewer Comments

Clo<u>s</u>e

Dear Scott,

We have fulfilled the requested revisions to the manuscript. We would like to thank the GigaScience editorial team and the reviewers for their constructive feedback and the opportunity for publishing our work in GigaScience.

On behalf of all authors, Björn

Clo<u>s</u>e